Author's response to reviews

Title: How much do the physician review and InterVA model agree in assigning probable causes of death? A Comparative analysis from rural Ethiopia

Authors:

Berhe Weldearegawi (berheph@gmail.com)
Yohannes Adama Melaku (yelidya@yahoo.com)
Geert Jan Dinant (beletejon@gmail.com)
Mark Spigt (berhe.sahle@monash.com)

Version: 3
Date: 24 June 2015

Author's response to reviews:

24 June 2015
MS: 1313637346164905
Title: How much do the physician review and InterVA model agree in determining causes of death? a comparative analysis of deaths in rural Ethiopia
Authors: Weldearegawi et al.
Dear Mr Rajendra Karkee
Journal Editorial Office
BioMed Central
Re: Submission of revised manuscript
Thank you for your letter dated 22nd Jun 2015, in which we have received feedback on the above-mentioned manuscript. We appreciate the thoughtful comments which were very helpful for improving the formatting of manuscript. We agree on the comments and we have revised our manuscript accordingly.
Appended to this letter is our point-by-point response to the comments raised by the editor(s).
Kind regards
Berhe Weldearegawi (on behalf-of the authors)

Comments and revisions made

1) Please place your tables in the main manuscript file after the figure legends and references. Please remove the copy from the additional files.
Response:
Thank you for the comments. The tables have now moved to the main manuscript, next to figure legends, and also remove d from the additional files.
2) Please remove the copies of the manuscript as additional files.
Response:
In our previous submission, tables were uploaded in the “additional files section”. As per the suggestions, they have been removed from this section and incorporated into the main manuscript.